Author's response to reviews

Title: The influence of the cultural climate of the training environment on physicians' self-perception of competence and preparedness for practice.

Authors:

Jamiu O Busari (j.busari@atriummc.nl)
Eduard AA Verhagen (fdmuski@yahoo.com)
Fred D Muskiet (a.a.e.verhagen@bkk.umcg.nl)

Version: 10 Date: 14 November 2008

Author's response to reviews:

Formatting changes requested
---------------------------------

References - The reference list should contain all authors' names, if there are 30 or below. The term et al. should not be used.

- Changes have been made as suggested

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Thank you.